Author's response to reviews

Title: Word Processing differences between dyslexic and control children

Authors:

Isabella C Paul (Isabella.Paul@uni-konstanz.de)
Christof Bott (Christof.Bott@uni-konstanz.de)
Christian Wienbruch (Christian.Wienbruch@uni-konstanz.de)
Thomas R Elbert (Thomas.Elbert@uni-konstanz.de)

Version: 4 Date: 22 November 2005

Author's response to reviews: see over
Dear Professor Newmark,

Thank you for accepting our manuscript for publication in BMC Psychiatry. We have now added information about ethical approval in the acknowledgement section. We further detailed how informed consent was given in the methods section.

Thank you again,
Sincerely yours,
Isabella Paul